Author's response to reviews

Title: Dynamic cardiac dyssynchrony is strongly associated with 2-year dialysis adequacy in continuous ambulatory peritoneal dialysis patients

Authors: 
Ching-Hui Huang (chinghui@cch.org.tw)
Chia-Chu Chang (chiachu@cch.org.tw)
Tzu-Lan Chang (tzulanchang@ntu.edu.tw)
Yu-Jun Chang (yujun@cch.org.tw)

Version: 4 Date: 14 March 2013

Author's response to reviews: see over
Dear Dr. Hayley Henderson,

It is our great honor to publish the paper in the high impact journal, BMC nephrology.

I have included the full name of the ethical committee, “Institutional Review Board and the Ethics Committee of the Changhua Christian Hospital, Taiwan, that granted approval for our study. (Page 6: Sentence 108 -109)

The authors also ensure that our revised manuscript conforms to the journal style.

Thank you again for acceptance of this manuscript.

Sincerely Yours

Chia-Chu Chang, MD
Division of Nephrology, Department of Internal Medicine
Changhua Christian Hospital, Changhua, Taiwan